Author's response to reviews

Title: Prevalence of physical, sexual and mental abuse among overweight adolescents.

Authors:

Jorien Veldwijk (jorien.veldwijk@rivm.nl)
Karin I Proper (karin.proper@rivm.nl)
Henriette B Hoeven-Mulder (h.hoeven@ggdgelre-ijssel.nl)
Wanda J.E Bemelmans (wanda.bemelmans@rivm.nl)

Version: 3 Date: 2 April 2012

Author's response to reviews:

Both the ethics committee of TNO-GP as well as the Dutch National Ethics Board (Central Committee on Research involving Human Subjects) concluded that formal testing by a medical ethical committee was not necessary as adolescents only had to complete a anonymous questionnaire once, which is in accordance with the guidelines laid down in the Declaration of Helsinki.